Author’s response to reviews

Title: Another Whipple’s Triad? Pericardial, Myocardial and Valvular Disease in an Unusual Case Presentation from a Canadian Perspective

Authors:

Christina Thornton (christina.thornton@ahs.ca)
Yinong Wang (Yinong.Wang@cls.ab.ca)
Martin Köbel (Martin.Koebel@cls.ab.ca)
Kathryn Bernard (kathy.bernard@canada.ca)
Tamara Burdz (tamara.burdz@canada.ca)
Andrew Maitland (Andrew.Maitland@albertahealthservices.ca)
Jose Ferraz (ferraz@ucalgary.ca)
Paul Beck (plbeck@ucalgary.ca)
Andre Ferland (Andre.Ferland@albertahealthservices.ca)

Version: 3 Date: 11 Nov 2019

Author’s response to reviews:

1. Title page

The email address of the corresponding author still differs between the title page and the submission system. Please ensure they match.

-This has been ensured to be correct to the same email of christina.thornton@ahs.ca

2. Ethics approval and consent to participate statement.

Currently this statement is unsuitable. Please replace this statement with 'Not applicable'.

-This has been replaced on line 221.
3. CARE checklist

Please remove the CARE checklist as it is no longer necessary at this stage.

-This has been removed.